Author's response to reviews

Title: What works in practice? User and provider perspectives on the acceptability, affordability, implementation, and impact of a family-based intervention for child overweight and obesity delivered at scale

Authors:

Patricia J Lucas (patricia.lucas@bristol.ac.uk)
Katherine Curtis-Tyler (katherine.tyler.1@city.ac.uk)
Lisa Arai (l.arai@tees.ac.uk)
Sally Stapley (s.stapley@tees.ac.uk)
Jamie Fagg (j.fagg@ucl.ac.uk)
Helen M Roberts (h.roberts@ucl.ac.uk)

Version: 3
Date: 15 May 2014

Author's response to reviews:

Dear Editor

Thank you to our reviewer for final comments. We have amended the two sentences in questions (changes in capitals). They now read:

"Parents and children were asked FOR and provided written consent with the exception of ONE YOUNG CHILD who gave verbal consent."

and

"For some, retaining half the families in the programme was considered sufficient, while other thought this was “appalling”"

Many thanks

Patricia Lucas